CENWP-OP




JULY 17, 2003

MEMORANDUM FOR THE RECORD

SUBJECT:  DRAFT Minutes for the July 16, 2003 FPOM Meeting

1.  The meeting was held in the Portland District office.  In attendance were:

Rex Baxter

USACE-WW

509.527.7124

Dave Hurson

COE-WW

509.527.7125

Scott Boyd

COE-NWD

503.808.3943

Steven Richards

WDFW


509.545.2050

Steve Haeseker

USFWS


360.696.7605

Tim Darland

COE-Bonn.

541.374.4551

Rick Martinson

PSMFC


541.296.8989

Bob Cordie

COE-TD/JD

541.298.7406

Miro Zyndol

COE-TD/JD

541.506.7823

Gary Johnson

COE-NWP

503.808.4304

Gary Fredricks

NMFS


503.231.6855

Scott Bettin

BPA


503.230.4573

Cal Sprague

COE-NWP

503.808.4305

2.  The following handouts were distributed:

· AGENDA, FISH PASSAGE O&M COORDINATION TEAM, PORTLAND DISTRICT OFFICE, ROOM 3h, July 16, 2003 (1300-1600)

· CENWP-OP-B, 16 July 2003, MEMORANDUM FOR: FPOM MEMBERS, SUBJECT: Bonneville Lock and Dam Fishway and construction Activities Report for July 16, 2003 FPOM meeting.

· Date: 7/16/03, From: Bob Cordie, Project Biologist, The Dalles Dam/John Day Dam, Re; FISH PASSAGE OPERATION AND MAINTENANCE MEETING CONSTRUCTION STATUS/FISHWAY STATUS.
· THE DALLES/JOHN DAY PROJECT, FISH RESEARCH REPORT, 7/15/2003
· -DRAFT #3-, THE DALLES DAM FISHWAY DEWATER/DIVE SCHEDULE, WINTER ‘03/’04

· -DRAFT #3-, JOHN DAY DAM FISHWAY DEWATER/DIVE SCHEDULE, WINTER ‘03/’04

· Email to Sprague, Calvin R NWP from Stansell, Robert J NWP detailing three questions on how to proceed with fish count website reporting for COE and PUD dams.

· CHINOOK FALLBACK COUNTS DURING SHAD FISHING ACTIVITY – THE DALLES EAST LADDER

· (River Flow Forecasts) 2003 summary, July Final

· Technical Report 2003-04, Effects of the Shad Fishery on Passage of Adult Chinook Salmon through the Oregon-shore Fishway Ladder at The Dalles Dam – 2002
· DRAFT, STATE OF WASHINGTON, DEPARTMENT OF FISH AND WILDLIFE FISH PROGRAM, Snake River Lab., … July 15, 2003 (letter to Wayne John) concerning Lower Granite adult trap.

3.  Construction Updates: Bonn.: (handout): Tim Darland discussed the handout entries.  B2CC construction is way ahead of schedule.  There are two new items: 1) 12 new headgates were delivered Monday (14th), and unit 9 was used as a staging area for their installation; 2) Phase 2 crane rail replacement begins August 1, ’03.

TD/JD: Miro reported that there are no new construction items at their project.  TD spill basin wall contract is being released.  Tom Lorz asked about repair of avian lines at the TD sluiceway outfall (they were vandalized about a month ago).  Bob Cordie said they will be replaced next spring.  Bob also said gulls are getting under the avian lines that are suspended from The Dalles bridge.  They have learned to go under the bridge first to get there.  They are susceptible to pyrotechnics from the top of the dam and leave readily when “treated” to them.  Gary F. said he is concerned about the sluiceway lines not being replaced.  Bob C. said if we push for replacing the lines this year, there will be many coordination and funding issues to resolve, delaying the process.  So, Gary F. said to keep Wildlife Services in the area as much as feasible and pyro/haze/take birds (i.e. protect this area) if we do not replace wires this year.

NWW: Rex discussed the handout, and will address agenda item 7 then, not now.

4. Fishway Status: Bonn.  Tim discussed the handout items.  P2 units out of service for annual maintenance.  FU 1 is OOS for vibration testing.  AFF picket lead #2 was OOS in the down position for a period of time, but it has been fixed.  Gary F. said (when on the project) that he saw the project personnel cleaning something in the fishway at weir 37.  No one (at this meeting) knows what they were doing.

TD/JD (highlights): Gull predation is occurring under some of the avian lines (bridge) and the project is working the pyros there.  At John Day, three pumps on the north entrance are back in service, and there is a 4th available if needed.  The project is working on getting a fifth pump available.  

JD Smolt Mon. Facility.: Facility was dewatered, and fish numbers were reported in the Status report.  Larry Basham asked how long it took, and the answer was 3-4 hours.  Tom Lorz asked what the water temps. were, and the answer was 65 – 69F.  Project reported that the STS camera failed during the summer screen inspection (trolley guide system started smoking), so the project is pulling screens instead.  

Project reported that the JD outfall avian cannon pump failed.  Their consensus was “do not replace it”.  Gary F. said “Take it out”.  For now, there will be no operating avian cannon at JD outfall.  The wires are in place, and are protecting the area from avian predation.  Tom L. said that if the wires come down, we will have to revisit this topic.  Gary F. said to make sure the wires stay up.  Bob said the pole that the wires are attached to is greased and wires are out of reach.

Bob C. reported that the John Day River riparian zone is being planted with trees by COE employees to exclude cattle from the riparian zone.  Rock Creek (further upriver on the WA side) is on the list for similar work.  ACTION: Gary Johnson asked Dave H. to send our Office of Council the records on this.  Dave said they are sending a letter to the cattle company and Gary J. said he wants to be cc’d on it.

NWW: Rex discussed handout.  He said that fish transport has been pretty routine with a few exceptions: 1) July 10: Barged fish were released at the JD forebay close to the start of spill at 1900 hrs. due to navigation lock closures/repairs taking place then.  2) At Lower Granite, high mortality rates were seen at the juvenile monitoring facility (3-5%), and that it is not usually this high.  3) Last Sunday (July 13), Mike Halter from Lower Granite called and said 700 of 4500 fish were morts, but that the descaling rates were fine.  It may have been a result of a temperature differential created by releasing water from Dworshak, creating stratification in the reservoir adjacent to the dam and sampling collector.  The lower layer stays cool, and the gradient may be compounded by running units at high end of the 1% range, so he called Rudd and they started running units at the lower end of 1%.  The problem was significantly lower from 10% morts to 0.8% morts and 0.1% morts the next day.  They have 3 units (MUs 2-4) operating.  Rex will get hourly data and correlate it to lowered morts seen.  4) Powerhouse loading at McNary: Increased morts seen.  Switch to north end of the project.  Temperatures are lower there.  We can go to the north end units but there may be an increase in predation.  Consensus of the group is: monitor the situation but with south end units off, an eddy forms.  We have new 1% numbers so we can’t do it anyway. They want those units back on line.  (Rex/Dave, please review this #4 and correct if necessary).  5) Ice Harbor oil leak: an estimated 450 gallons of oil were lost, so they closed the orifices in the unit.  Apparently none of the oil got in the water outside of the dam.  No other problems to report.

5. Research Updates: Bonn.: Battelle is doing a gap-loss study.  Thursday (last) ended the test with the TIES in, so now they will pull TIES on MU 15 and an adjacent unit and continue the gap loss test.

TD/JD: Miro reported that not much research was taking place and that most of the tests were completed. (see handout).

NWW: Turbine survival study taking place at Lower Granite.  The RSW test will involve a drop-down deployment (summer turbine study).

The transport study at McNary is almost completed.  Very few steelhead were found for the study.  Most were diverted.  They are looking at 1000-2000 juveniles and 10 adults.  So this raises the question on whether they can do the steelhead portion of the study.  Scott Boyd stated that the turbine study leaves the units within 1% as usual.

6.  Task Groups:  Count Program:  There was a discussion on the email handout from Robert Stansell on whether to have a link to PUD counts or not to have a link  (to their website).  ACTION:  More in-house discussions to take place to hash out the details.  I will send out an email soliciting availability for a conference call to be held next week (week of July 21).   Scott Boyd said he felt that the way we report PUD counts now is still workable and that a link may create discrepancies in methods of retrieval of data.

Shad Fishery at TDA :  ACTION: I am to see if I have a copy of the section 10 permit from the origins of the fishery.  I was also tasked with sending out the 2003 Shad Fishery report (I did that yesterday, July 17) to FPOM members via email.  I will also try to contact Robert Bailey for information.  Gary Fredricks said we should table this discussion until the next FPOM meeting in August.  In the meantime I will find out what I can on the triggers to discontinue the fishery (i.e. the frequency of fallback events that occur over 10% that are triggers to terminate or modify the fishery).

(New Task Group): Dalles Sluiceway Operations Task Group: Chair is BPA’s Scott Bettin.  Scott requested a decision to be made on a proposal to shut the orifices in the gatewells at TDA, and open a chain-gate into the sluiceway to let in the same amount of water, as this may help pass fish from the forebay more efficiently.  This is not a request for a benefit from the power standpoint.  It is about whether more fish-laden water can be passed through the chaingates into the sluiceway.  Is the method used now the most efficient method to pass juveniles?  ACTION:  Scott will send out an email on this topic with information on the request.

7.  Lamprey Protection During Dewaterings:  TDA is going to build some device to keep lamprey from entering the diffuser gratings when the fishway is dewatered.  They want to maintain flushing water, provide an extra recovery tank to keep up with the large number of rescued lamprey, and lastly, the project will NOT be entering the diffuser chambers to recover lamprey, so some other method must be derived.  Gary J. said to re-convene the task group, so, ACTION: I will send out task group attendance info.  I will also set up a conference call for the Lamprey Task Group to brainstorm ideas and discuss them.  The group will need time to put together a bulleted list according to Dave H.  Bob C. said the JD north diffuser area at the countstation has been a place where old, decayed lamprey morts have been seen.  In response, Bob proposed configuring a grating with smaller openings to prevent lamprey from entering the diffuser chamber.

8.  Ice Harbor Fishway Outages:  Dave Hurson sent out an email message in June, and again described the problem here.  The project will be rehabbing 8 pumps this winter.  The pumps can’t be unwatered by themselves.  There will be 8 sets of guides for bulkheads.  There will be a 2-month outage on the North Shore, with a 2 week concurrent outage of both fishways.  Pumps 7, 8, and 9 may be OOS during March (will operate 6 pumps).  It may take 3 months to get all the butterfly valves done, but should be completed by April 1, ‘04.  There was a long discussion between NMFS, FPC and Dave.  The COE unwaters the facility, and the contractor does the rest of the work.  Larry B. and Gary F. both said that the sooner the COE/contractor finishes the south shore fishway, the better.

Steve Richards asked if the COE can replace the count station window, since it is nearly “opaque”, and the COE answered, “Probably”.

9.  Ice Harbor Fish Pump Bulkheads.  Dave explained the schedule (this write-up is in the Construction Updates handout). 

10.  JDA Gatewell Orifice Closure During WMP:  Topic for next FPP revision process.  Scott B. (BPA) is requesting closure of the orifices to save revenue by saving water.  He estimated the $$ amount to be around $1Mil in savings.  The FPOM FPP revision process will address this proposal.  At first, it was decided to keep 1/3 of the units’ gatewell orifices open, and cycle them Mon/Wed/Fri.  Rex then proposed an alternative:  Close all the orifices on Dec. 16th.  Dip gatewells to see if we have steelhead.  (This idea did not seem totally accepted by the FPOM group).  ACTION: The consensus was to table this topic until next FPOM meeting at which time we’ll proceed with the discussion.

11.  Updates for the FPP: Dave H. said that ACTION: Before the next FPOM meeting, we (Dave and I) will get a bulletized list together for distribution.  By November, we hope to have revisions in draft form.  Scott requested we send out teletypes of FPP revisions to FPOM after they are finalized, so everyone knows what’s being done, when its being done.  

Miro stated that there is no Unit Operating Priority at TDA (I recalled there being one, coordinated through the May FPOM with an email sent out for two week review, and then a teletype was sent to the project lining out the unit operating priority (the one proposed by Dick Harrison).  Therefore, I think there IS a unit operating priority in place at TDA.  (Scott Boyd then reaffirmed my suspicions when he said that the teletype went out on May 28 following a two-week comment period following an email I sent requesting comments but received none).

12.  River Flow Forecast: (handout): Scott Boyd went through handout.  Operations for Libby are being debated, and for Dworshak are on hold.  Projections are for Lower granite to be lower (than originally forcasted), and for McNary to be higher.  

13.  Other:  

· TDA Main Units 1-4 operation proposal (due to Pendleton Wind Farm being on the same transmission line as TDA):  Scott Bettin outlined two options: 1) 2 units run at higher end of 1%, or 2) all 4 units run at low end of 1%.  (The consensus was): run MU 1 and 2 at higher load.  

· Rex B. handed out a letter from WDFW.  The sampling personnel want to operate LG trap at 70 – 72F.  ACTION: Rex will email and ask for comments for discussion next week. Therefore, read the letter and provide feedback

14.  Next Meeting:  August 14, 2003 at Robert Duncan Plaza, 0900 – 1300 hours, room TBA

